Author's response to reviews

Title: A positioning pillow to improve lumbar puncture in paediatric haematology-oncology patients: A randomized controlled trial.

Authors:

Perrine Marec-Bérard (marec@lyon.fnclcc.fr)
Alvine Bissery (alvine.bissery@chu-lyon.fr)
Kamila Kebaïli (kamila.kebaill@chu-lyon.fr)
Matthias Schell (schell@lyon.fnclcc.fr)
Françoise Aubert (francoise.aubert@chu-lyon.fr)
Ségolène Gaillard (seglene.gaillard@chu-lyon.fr)
Muriel Rabilloud (muriel.rabilloud@chu-lyon.fr)
Behrouz Kassaï (bk@upcl.univ-lyon1.fr)
Catherine Cornu (catherine.cornu@chu-lyon.fr)

Version: 4 Date: 6 January 2009

Author's response to reviews: see over
Dear Melissa Norton, Editor-in-Chief of BMC Cancer

We are happy to submit our revised manuscript entitled:

"A positioning pillow to improve lumbar puncture success rate in paediatric haematology-oncology patients: A randomized controlled trial."

We addressed the reviewer comments and added the following sentence in the discussion section (e.g. first paragraph, lines 6 & 7):

"There was no statistically significant difference between LP rate of success with and without pillow. Results were also indeterminate for all secondary criteria."

We are hopeful that this version of our manuscript is acceptable for publication in BMC Cancer.

Sincerely yours,

P Marec-Bérard
Corresponding author.